Author’s response to reviews

Title: Nitrous oxide occupational exposure in conscious sedation procedures in pediatric dental ambulatories: a pilot retrospective observational study in an Italian pediatric hospital

Authors:
Salvatore Zaffina (salvatore.zaffina@opbg.net)
Marco Lembo (marco.lembo@opbg.net)
Francesco Gilardi (francesco.gilardi@opbg.net)
Andrea Bussu (andrea.bussu@opbg.net)
Fabio Pattavina (fabio.pattavina@policlinicogemelli.it)
Graziano Tucci (graziano.tucci@opbg.net)
Umberto Moscato (umberto.moscato@unicatt.it)
Massimiliano Raponi (massimiliano.raponi@opbg.net)
Pietro Derrico (pietro.derrico@opbg.net)
Angela Galeotti (angela.galeotti@opbg.net)
Vincenzo Camisa (vincenzo.camisa@opbg.net)

Version: 3 Date: 29 Dec 2018

Author’s response to reviews:

To The Editor
BMC Anesthesiology

Dear Editor,

as requested, we made major revisions to our article “Nitrous oxide occupational exposure in conscious sedation procedures in pediatric dental ambulatories”.

(Reviewer 3):
Thank you for the comments to our manuscript.

We did our best to improve our manuscript and hope that this version meets your requirements.

Therefore, according to your requests, we:

• revised our article according to the instructions for the authors and STROBE guidelines
• better explained the evidence level of the study
• checked and changed the trade names where necessary
• controlled and corrected all the abbreviations
• inserted the tables at the end of the document text file
• implemented the tables explanations
• corrected the distinction of "." and "," in tables
• emphasized the novelty of the study

Looking forward to hearing from you, I thank you for your kind attention and send you my best regards.

Bambino Gesù Children's Hospital, IRCCS

Health Directorate
Head of Occupational Health
Dr. Salvatore Zaffina